Author's response to reviews

Title: Study protocol: Determinants of participation and quality of life of adolescents with cerebral palsy: a longitudinal study (SPARCLE2)

Authors:

Allan F Colver (allan.colver@ncl.ac.uk)
Heather O Dickinson (heather.dickinson@ncl.ac.uk)

Version: 6 Date: 5 May 2010

Author's response to reviews:

Dear Sir

I have addressed the revisions as follows

Major revisions (we require the author to make these changes)

Author groups: please include the name of the group within the list of authors, by adding the name to the `first name' box; and the bracketed abbreviation to the `last name' box in the submission system. DONE

Please remove the names of the members of the group from the title page and place them in the Acknowledgements, as they are to be acknowledged as members of the study group.

The groups' contribution as a whole can be given in the authors contributions section. DONE

Table 1: This table had been divided into two parts, please merge these into one table. DONE

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Abstract: Please remove the word count. DONE

Appendix: Please move this to after the Authors' contribution. DONE

Figure: Please refer to this as 'Figure 1' throughout the manuscript. DONE

Tables- please remove the visible vertical lines from your tables DONE

Yours sincerely

Allan Colver